Author's response to reviews

Title: Molecular breast cancer subtypes and therapies in a public hospital in Northeastern Brazil

Authors:

Ana Cláudia Macedo Andrade (anaclaudia2008@hotmail.com)
Carlos Alberis Ferreira Júnior (alberis_aj@hotmail.com)
Beatriz Dantas Guimarães (biadantasq@gmail.com)
Ana Waleska Pessoa Barros (waleskabarros2011@gmail.com)
Gibran Sarmento de Almeida (gibransarmento@hotmail.com)
Mathias Weller (mathiasweller@uepb.edu.br)

Version: 3
Date: 25 August 2014

Author's response to reviews: see over
Dear editor,

According to STROBE guidelines (Method-Setting-5) the sentence “Data were sampled between March and November 2013” (line 125) was added. Furthermore, the whole manuscript was checked and adheres to STROBE guidelines.

A list of abbreviations was added (line 374-376):

“List of abbreviations used”

“ER: estrogen receptor; HR: hormone receptor; PR: progesterone receptor; TN: triple negative.”

Kind regards. Mathias W.